EAS 305 Quiz 4, Spring 1998 Page 1 of 1

Answer the following problems. (5 points eac h), 60 minutes.

1- Given data points . find coefficients of linear expression , such that the square error is minimized in the following two cases: a- Three data points b- Four data points

2- Continuous Random variables and have the joint pdf:

Find the linear regression of on .

m x1 y1,( ) x2 y2,( ) … xm ym,( ) , , , a0 a1, ŷ x( ) a0 a1x+=

1 1,( ) 2 3,( ) 3 2,( ) , , 1 1,( ) 1 2,( ) 3 2,( ) 3 4,( ) , , ,

X Y

fX Y, x y,( ) 1 2 -- 2xy 0 x 1 0 y 1≤ ≤ , ≤ ≤ ; + 0 Otherwise ; =

y x